Author's response to reviews

Title: The effectiveness of interventions targeting specific out-of-home food outlets: protocol for a systematic review

Authors:

Frances C Hillier-Brown (frances.hillier@durham.ac.uk)
Helen J Moore (helen.moore@durham.ac.uk)
Amelia A Lake (amelia.lake@durham.ac.uk)
Ashley J Adamson (ashley.adamson@newcastle.ac.uk)
Martin White (martin.white@newcastle.ac.uk)
Jean Adams (jean.adams@newcastle.ac.uk)
Vera Araujo-Soares (vera.araujo-soares@newcastle.ac.uk)
Charles Abraham (C.Abraham@exeter.ac.uk)
Carolyn D Summerbell (carolyn.summerbell@durham.ac.uk)

Version: 2 Date: 10 January 2014

Author's response to reviews:

Added reference (Riley 1994) that was missing (page 6)
Deleted databases that we are not searching (pages 11-12)